////
Automatically generated by PainlessDocGenerator. Do not edit.
Rebuild by running `gradle generatePainlessApi`.
////

[[painless-api-reference-ZoneOffsetTransition]]++ZoneOffsetTransition++::
* ++[[painless-api-reference-ZoneOffsetTransition-of-3]]static <<painless-api-reference-ZoneOffsetTransition,ZoneOffsetTransition>> link:{java8-javadoc}/java/time/zone/ZoneOffsetTransition.html#of%2Djava.time.LocalDateTime%2Djava.time.ZoneOffset%2Djava.time.ZoneOffset%2D[of](<<painless-api-reference-LocalDateTime,LocalDateTime>>, <<painless-api-reference-ZoneOffset,ZoneOffset>>, <<painless-api-reference-ZoneOffset,ZoneOffset>>)++ (link:{java9-javadoc}/java/time/zone/ZoneOffsetTransition.html#of%2Djava.time.LocalDateTime%2Djava.time.ZoneOffset%2Djava.time.ZoneOffset%2D[java 9])
* ++[[painless-api-reference-ZoneOffsetTransition-compareTo-1]]int link:{java8-javadoc}/java/time/zone/ZoneOffsetTransition.html#compareTo%2Djava.time.zone.ZoneOffsetTransition%2D[compareTo](<<painless-api-reference-ZoneOffsetTransition,ZoneOffsetTransition>>)++ (link:{java9-javadoc}/java/time/zone/ZoneOffsetTransition.html#compareTo%2Djava.time.zone.ZoneOffsetTransition%2D[java 9])
* ++[[painless-api-reference-ZoneOffsetTransition-getDateTimeAfter-0]]<<painless-api-reference-LocalDateTime,LocalDateTime>> link:{java8-javadoc}/java/time/zone/ZoneOffsetTransition.html#getDateTimeAfter%2D%2D[getDateTimeAfter]()++ (link:{java9-javadoc}/java/time/zone/ZoneOffsetTransition.html#getDateTimeAfter%2D%2D[java 9])
* ++[[painless-api-reference-ZoneOffsetTransition-getDateTimeBefore-0]]<<painless-api-reference-LocalDateTime,LocalDateTime>> link:{java8-javadoc}/java/time/zone/ZoneOffsetTransition.html#getDateTimeBefore%2D%2D[getDateTimeBefore]()++ (link:{java9-javadoc}/java/time/zone/ZoneOffsetTransition.html#getDateTimeBefore%2D%2D[java 9])
* ++[[painless-api-reference-ZoneOffsetTransition-getDuration-0]]<<painless-api-reference-Duration,Duration>> link:{java8-javadoc}/java/time/zone/ZoneOffsetTransition.html#getDuration%2D%2D[getDuration]()++ (link:{java9-javadoc}/java/time/zone/ZoneOffsetTransition.html#getDuration%2D%2D[java 9])
* ++[[painless-api-reference-ZoneOffsetTransition-getInstant-0]]<<painless-api-reference-Instant,Instant>> link:{java8-javadoc}/java/time/zone/ZoneOffsetTransition.html#getInstant%2D%2D[getInstant]()++ (link:{java9-javadoc}/java/time/zone/ZoneOffsetTransition.html#getInstant%2D%2D[java 9])
* ++[[painless-api-reference-ZoneOffsetTransition-getOffsetAfter-0]]<<painless-api-reference-ZoneOffset,ZoneOffset>> link:{java8-javadoc}/java/time/zone/ZoneOffsetTransition.html#getOffsetAfter%2D%2D[getOffsetAfter]()++ (link:{java9-javadoc}/java/time/zone/ZoneOffsetTransition.html#getOffsetAfter%2D%2D[java 9])
* ++[[painless-api-reference-ZoneOffsetTransition-getOffsetBefore-0]]<<painless-api-reference-ZoneOffset,ZoneOffset>> link:{java8-javadoc}/java/time/zone/ZoneOffsetTransition.html#getOffsetBefore%2D%2D[getOffsetBefore]()++ (link:{java9-javadoc}/java/time/zone/ZoneOffsetTransition.html#getOffsetBefore%2D%2D[java 9])
* ++[[painless-api-reference-ZoneOffsetTransition-isGap-0]]boolean link:{java8-javadoc}/java/time/zone/ZoneOffsetTransition.html#isGap%2D%2D[isGap]()++ (link:{java9-javadoc}/java/time/zone/ZoneOffsetTransition.html#isGap%2D%2D[java 9])
* ++[[painless-api-reference-ZoneOffsetTransition-isOverlap-0]]boolean link:{java8-javadoc}/java/time/zone/ZoneOffsetTransition.html#isOverlap%2D%2D[isOverlap]()++ (link:{java9-javadoc}/java/time/zone/ZoneOffsetTransition.html#isOverlap%2D%2D[java 9])
* ++[[painless-api-reference-ZoneOffsetTransition-isValidOffset-1]]boolean link:{java8-javadoc}/java/time/zone/ZoneOffsetTransition.html#isValidOffset%2Djava.time.ZoneOffset%2D[isValidOffset](<<painless-api-reference-ZoneOffset,ZoneOffset>>)++ (link:{java9-javadoc}/java/time/zone/ZoneOffsetTransition.html#isValidOffset%2Djava.time.ZoneOffset%2D[java 9])
* ++[[painless-api-reference-ZoneOffsetTransition-toEpochSecond-0]]long link:{java8-javadoc}/java/time/zone/ZoneOffsetTransition.html#toEpochSecond%2D%2D[toEpochSecond]()++ (link:{java9-javadoc}/java/time/zone/ZoneOffsetTransition.html#toEpochSecond%2D%2D[java 9])
* Inherits methods from ++<<painless-api-reference-Object,Object>>++
